End of Term Announcements

Print Name:______________________________

ID: ______________

Course Evaluations

• Write your name and ID above and complete the attached course evaluation. • Submit this form and the evaluation at the final exam. Separate submission boxes will be used to guarantee anonymity.

• To help ensure a 100% response rate, five points will be deducted from your final exam score if we do not have this sheet with your signature and ID.

Preparing for the Final Exam

• The final exam will be based on S1-S12, P1-P7, Pre1-Pre3, and L1-L28. See the website. • Review session on Tuesday, May 11, 7:30-9:00pm, Olin 155. Study week office hours:

Monday Tuesday Wednesday

10:00-11:30 DK 5143 Ups 10:00-11:30 DK 5143 Ups 9:30-11:00 KH 4120 Ups 10:30-12:00 KH 4120 Ups 1:00-2:30 YY 393 Rhodes 10:00-12:00 CV 5146 Ups 12:00-2:00 MA 334 Ups 2:00-3:30 HN 326 Ups 1:00-2:30 YY 393 Rhodes 1:30-3:00 KW 4156 Ups 2:30-4:00 BB 4114 Ups 2:30-4:00 BB 4114 Ups 2:00-3:30 HN 326 Ups 3:00-5:00 CV 5146 Ups 3:00-4:30 KW 4156 Ups 3:00-4:30 AB 4162 Ups 3:00-4:30 AB 4162 Ups 3:30-5:00 CV 5146 Ups

The Final Exam

• The final exam is Thursday, May 13, 9:00AM-11:30AM, Barton Hall. • You must bring your Cornell ID as it will be checked. • If you have a valid reason for taking the exam at an alternate time, contact buck@cs.cornell.edu by Friday, May 7.

Grades Etc.

• P7 solutions on web Monday. Carpenter P7 pick-up after Wednesday noon. Submit regrades at the final.

• Final exam solutions and results on web Wednesday 5/19. Course grades posted in Carpenter Wednesday 5/19.

• If you want your grade and final exam score by postcard, submit a stamped postcard at the final.

You can look at your exam in Upson 303 during June 1 � August 1 or after the third week of classes during the fall semester.